Author’s response to reviews

Title: Trial and error: Evaluating and refining a community model of HIV testing in Australia

Authors:

Kathleen Ryan (kathleen.ryan@burnet.edu.au)
Alisa Pedrana (alisa.pedrana@burnet.edu.au)
David Leitinger (david.leitinger@pronto.org.au)
Anna Wilkinson (anna.wilkinson@burnet.edu.au)
Peter Locke (peter.locke@vac.org.au)
Margaret Hellard (margaret.hellard@burnet.edu.au)
Mark Stoove (mark.stoove@burnet.edu.au)

Version: 2 Date: 20 Sep 2017

Author’s response to reviews:

Technical Comments:

1) Please include your Abstract in your manuscript before the Background.

The abstract has now been included in the manuscript

2) Please change the title of the "Competing interesting" section to "Competing interests"

"Competing interests" is now spelt correctly.